

https://doi.org/10.1093/pnasnexus/pgac219 Advance access publication date: 3 October 2022 Research Report

Socioeconomic disadvantage, chronic stress, and proinflammatory phenotype: an integrative data analysis across the lifecourse

Phoebe H. Lam ^[]a,*, Edith Chen^{a,b}, Jessica J. Chiang^c and Gregory E. Miller^{a,b}

^aDepartment of Psychology, Northwestern University, Swift Hall, 2029 Sheridan Road, Evanston, IL 60208, USA

^bInstitute for Policy Research, Northwestern University, 2040 Sheridan Road, Evanston, IL 60208, USA

^cDepartment of Psychology, Georgetown University, 306 N White-Gravenor Hall, 37th and O Streets, NW, Washington DC, 20057, USA

*To whom correspondence should be addressed: Email: phoebelam@u.northwestern.edu

Edited By: Karen E. Nelson

Abstract

Socioeconomic disadvantage confers risk for many chronic illnesses, and theories have highlighted chronic psychological stress and alterations to inflammatory processes as key pathways. Specifically, disadvantage can heighten chronic stress, which may promote a proinflammatory phenotype characterized by immune cells mounting exaggerated cytokine responses to challenge and being less sensitive to inhibitory signals. Importantly, lifecourse perspectives emphasize that such immune alterations should be more potent earlier in life during a sensitive period when bodily tissues are highly plastic to environmental inputs. However, examining these propositions is resource intensive, as they require cell-culturing approaches to model functional inflammatory activities, a wide age range, and longitudinal data. Here, we integrated data from five independent studies to create a diverse sample of 1,607 individuals (960 with longitudinal data; 8 to 64 years old; 359 Asian, 205 Black, and 151 Latino/a). Leveraging the resulting lifecourse data, rich interview assessments of disadvantage and stress, and ex vivo assessments of inflammation, we examined two questions: (1) Does chronic stress account for the link between disadvantage and proinflammatory phenotype? (2) Is there a developmental period during which inflammatory responses are more sensitive to disadvantage and chronic stress? Disadvantage was associated with higher chronic stress, which was linked with a proinflammatory phenotype cross-sectionally, longitudinally, and in terms of prospective change across 1.5 to 2 years. Consistent with the sensitive period hypothesis, the magnitude of these indirect associations was strongest in earlier decades and declined across the lifecourse. These findings highlight the importance of taking a lifecourse perspective in examining health disparities.

Keywords: socioeconomic disadvantage, chronic psychological stress, inflammation, development, lifecourse

Significance Statement:

Chronic diseases are patterned by socioeconomic status (SES), but the underlying pathways and how their magnitude vary across the lifecourse are unclear. Here, in an integrated sample of 1,607 individuals aged 8 to 64 years, we found that socioeconomic disadvantage was associated with higher chronic stress, which in turn predicted a proinflammatory phenotype characterized by immune cells mounting aggressive responses to challenges and being less sensitive to inhibition signals. Importantly, the magnitude of these links was strongest earlier in life and declined across the lifecourse, supporting the notion of a sensitive developmental period during which environmental input may have stronger impacts. If replicated, these findings suggest that efforts to mitigate health disparities may be most efficacious when implemented during childhood.

Introduction

Socioeconomic disadvantage—typically defined by low income, savings, or education level—is associated with physical health problems throughout the lifecourse. In childhood and adolescence, disadvantage is associated with indicators of cardiovascular risk, such as higher blood pressure and fasting glucose; in adulthood, it is linked with morbidity and mortality from various chronic illnesses, including cardiovascular, respiratory, autoimmune diseases, and some cancers (1–6). These findings have spurred conceptual models attempting to explain how socioeconomic disadvantage contributes to such a wide range of health problems across the lifecourse, and excessive inflammation has been highlighted as one key pathway.

Inflammation is one of the body's primary defense mechanisms against invading pathogens and tissue damage. When such infections or injuries are detected, cells of the innate immune system—monocytes, dendritic cells, and macrophages—mount an acute inflammatory response, which is coordinated by proinflammatory cytokines, such as Interleukin (IL)-6, IL-1 β , and Tumor Necrosis Factor (TNF)- α . These proinflammatory cytokines attract other

Competing Interest: The authors declare no competing interest.

Received: June 3, 2022. Accepted: September 28, 2022

© The Author(s) 2022. Published by Oxford University Press on behalf of National Academy of Sciences. This is an Open Access article distributed under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs licence (https://creativecommons. org/licenses/by-nc-nd/4.0/), which permits non-commercial reproduction and distribution of the work, in any medium, provided the original work is not altered or transformed in any way, and that the work is properly cited. For commercial re-use, please contact journals.permissions@oup.com immune cells to the site, and cause those cells to proliferate, release antimicrobial signals, and mature into phenotypes best suited to countering the threat. The acute inflammatory response is critical for survival; however, it needs to be tightly regulated because sustained "nonresolving" inflammation can damage tissue and organs in ways that increase risk for a range of chronic diseases (7). This regulation is mediated via both local (e.g. IL-10) and systemic (e.g. cortisol) signals that, at sufficiently high concentrations, suppress the acute inflammatory response by inhibiting cell recruitment and cytokine production.

In the health disparities literature, there has been considerable interest in nonresolving inflammation as a mechanistic pathway. Typically, studies have measured inflammation by quantifying protein biomarkers like C-Reactive Protein (CRP), IL-6, and TNF- α in circulating blood. This approach is motivated by longitudinal studies indicating that circulating biomarkers forecast risks of morbidity and mortality from diabetes, heart attack, and stroke, as well as functional decline and premature death (8-10). Over the past two decades, numerous studies have considered whether concentrations of these biomarkers are heightened in socially disadvantaged individuals. Several meta-analyses have synthesized these results, concluding that lower socioeconomic status (SES) is associated with higher levels of circulating protein biomarkers, which are thought to reflect nonresolving inflammation (11–13). This association was observed in children, adolescents, and adults (11-13).

Despite the consistency of these results, they do not enhance scientific understanding of the mechanistic basis of the association between disadvantage and health. Cytokines such as IL-6 and TNF- α are released not only by innate immune cells, but also by endothelial, skeletal, and adipose cells, for purposes that do not always involve defending against infection or repairing tissue injuries (14–16). Similarly, CRP functions as an antimicrobial peptide, but the liver releases it when IL-6 is high, irrespective of the precipitating stimulus (17). Thus, these biomarkers do not necessarily reflect underlying inflammation. A more direct approach to measuring inflammatory processes is to utilize ex vivo methods that simulate an immune challenge. Specifically, studies have cultured immune cells with microbial stimuli, such as the bacterial product lipopolysaccharide (LPS), and then quantified the ensuing production of proinflammatory cytokines. Conceptually, this measure provides a proxy for the magnitude of the inflammatory response to challenges. In addition, cells can be simultaneously exposed to a microbial stimulus and a compound that exerts anti-inflammatory effects, such as cortisol or IL-10. The anti-inflammatory compound partially inhibits cytokine production, but the magnitude of this inhibition varies from person to person. As such, the extent of inhibition can be used as a proxy for cellular sensitivity to inhibitory signals. Conceptually, this measure represents the capacity of cells to respond to "stop" signals.

Some studies have considered whether these functional indicators of inflammatory activity vary as a function of SES. For example, individuals with low SES had immune cells that mounted larger proinflammatory responses following stimulation and had lower capacity to inhibit inflammatory responses (18, 19). However, these studies have relatively smaller sample sizes (N's = 37 to 150) and included few participants of color, limiting broad generalization of the results. Most of these studies also utilized crosssectional designs, so that inferences about directionality could not be made. The present study aimed to address these limitations by combining data from five independent studies, three of which utilized longitudinal designs, to carry out an integrative data analysis. Specifically, we extracted and harmonized measures of chronic stressors and ex vivo assessment of inflammatory processes to create a sample of over 1,600 individuals (over 900 with longitudinal data). The resulting sample spans a wide age range from 8 to 64 years and is diverse in race/ethnicity and SES, increasing the generalizability of results. Of note, this is achieved without the typical approach of relying on public data, for which there are increasing concerns about their overuse. Specifically, overuse of the same dataset creates dependencies among research papers that are falsely regarded as independent contributions as well as exacerbates sample-specific peculiarities and biases that may not replicate (20). Furthermore, previous studies have largely analyzed responses to stimulation and sensitivity to inhibition as independent outcomes, masking potential coupling patterns of inflammatory processes. Therefore, the current investigation estimates a latent proinflammatory phenotype profile characterized by both relatively higher cytokine responses to challenge and relatively lower sensitivity to inhibitory signals.

In addition, the current investigation addressed two unresolved questions. First, conceptual models have postulated that socioeconomic disadvantage may impact immune functioning through various pathways, including early infection, environmental pollution, health behaviors, and psychological stress (21-24). However, formal examinations of mediation scenarios have rarely been conducted. Leveraging the fact that each of the studies had rich interview data assessing chronic psychological stress, we focused on this potential mechanistic pathway and examined whether it could account for cross-sectional and longitudinal links between socioeconomic disadvantage and proinflammatory phenotype. We hypothesized such an indirect effect because disadvantage is known to heighten chronic stress. For example, disadvantage has been linked to exposure to violence, discrimination, and strained relationships (25-27). In turn, nascent research suggests that chronic stress may be associated with higher cytokine responses to challenge and lower sensitivity to inhibitory signals (18, 28, 29).

Second, leveraging the wide age range of the integrated data, we examined whether there may be a developmental period during which inflammatory processes are more sensitive to disadvantage and chronic stress. Numerous conceptual models postulate sensitive developmental periods in early stages of life during which bodily tissues have heightened plasticity, and thus are more sensitive to environmental input (30-33). Indeed, animal studies have found that lower maternal care during the first week of rodents' lives led to physiological alterations, which persisted in adulthood (34, 35). By contrast, lower maternal care during the second week of rodents' lives had relatively smaller effects on physiology, and lower maternal care during the third week of rodents' lives had no effects on physiology as youth and as adults (36). However, empirical tests of this proposition in humans are challenging because they require lifecourse data. Here, we leveraged the wide age range that resulted from the integrated data to address this issue directly. Based on findings from animal models, we hypothesized that the magnitude of stress-inflammation relationships would be strongest in earlier stages of life and decline with age. It is worth noting that the immune system matures and acquires memory throughout the lifespan (37, 38) and can still be modulated by environmental input at older ages (39), allowing a more thorough our hypothesis, with fewer concerns about ceiling effects.

Table 1. Descriptive statistics of sample characteristics by study (Analytical N = 1,607).

	Study A	Study B: children	Study B: parents	Study C	Study D	Study E
Sample size	277	261	261	151	349	308
Location	Chicago	Vancouver	Vancouver	Vancouver	Vancouver	Chicago
Year	2015-2017	2010-2012	2010-2012	2005-2007	2009-2012	2012-2017
Design	Longitudinal	Longitudinal	Longitudinal	Longitudinal	Cross-sectional	Cross-sectional
Age range	11–15	13–16	32–64	14–19	15-55	8–17
Age Mean (SD)	13.45 (0.62)	14.53 (1.07)	45.83 (5.50)	17.01 (1.35)	36.48 (1.78)	12.99 (2.50)
Race			· · · ·			. ,
White	76 (27%)	129 (49%)	157 (60%)	76 (48%)	264 (73%)	155 (50%)
Black	106 (38%)	12 (5%)	4 (2%)	2 (1%)	2 (1%)	79 (25%)
Asian	14 (5%)	94 (36%)	83 (32%)	68 (43%)	68 (19%)	31 (10%)
Latino/a	79 (28%)	14 (5%)	11 (4%)	0 (0%)	5 (1%)	42 (14%)
POC	5 (2%)	12 (5%)	5 (2%)	11 (7%)	21 (6%)	4 (1%)
Female	175 (62%)	125 (48%)	199 (76%)	157 (100%)	198 (55%)	138 (44%)
Adiposity	23.57 (6.10)	21.37 (3.70)	25.38 (4.61)	21.70 (2.69)	25.76 (5.86)	22.43 (5.69)
SES			()			()
Income	5.24 (2.18)	5.35 (1.83)	5.35 (1.83)	N/A	4.69 (1.80)	6.35 (2.07)
Savings	3.86 (2.55)	4.52 (2.48)	4.52 (2.48)	N/A	2.99 (1.82)	4.94 (2.74)
Education	3.53 (1.25)	3.98 (0.86)	3.98 (0.85)	3.73 (1.03)	3.36 (1.24)	3.36 (1.24)
Inflammatory para	ameters	· · · · ·	· · · ·	· · · · ·	· · · · ·	· · · ·
Cells	Whole blood	Whole blood	Whole blood	Whole blood	Whole blood	PBMC
Ligand for SCP	LPS, R848, HSP60	LPS	LPS	LPS	LPS	PIC, ODN
Ligand for SI	LPS	LPS	LPS	LPS	LPS	PIC
Inhibitor; dose	CRT; 2, IL-10; 3	CRT; 2	CRT; 2	CRT; 4	CRT; 3	CRT; 1
Cytokines	IL-6, IL-1 β , IL-8,	IL-6, IL-1 β , IL-8,	IL-6, IL-1 β , IL-8,	IL-6 in	Intracellular IL-6	IL-6, IL-1 <i>β</i> , IL-8,
	TNF- α in	and IL-10 in	and IL-10 in	supernatant	in CD14 + mono-	TNF-α in
	supernatant	supernatant	supernatant	*	cytes	supernatant
Assay method	Immunoassays	Immunoassays	Immunoassays	Immunoassays	Flow cytometry	Immunoassays

Note: Study B has relevant data for both children and parents, thus descriptive statistics are presented separately. Location refers to the city in which the study was conducted. Year refers to the years during which data was collected. Family income was coded in a 9-point scale: 1 = less than 5,000 to 74,999; 3 = 20,000 to 74,999; 5 = 50,000 to 74,999; 5 = 50,000 to 74,999; 7 = 100,000 to 149,999; 8 = 150,000 to 199,999; and 9 = 200,000 and higher. Family savings was coded in a 9-point scale: 1 = less than 5,000 to 29,999; 7 = 100,000 to 19,999; 8 = 200,000 to 49,999; 6 = 50,000 to 99,999; 7 = 100,000 to 19,999; 5 = 20,000 to 49,999; 6 = 50,000 to 99,999; 7 = 100,000 to 199,999; 8 = 200,000 to 499,999; and 9 = 500,000 and higher. POC refers to other people of color. Adiposity was assessed with body mass index. PBMC refers to peripheral blood mononuclear cell. LPS refers to lipopolysaccharide. R848 refers to Resignimod. HSP60 refers to heat shock protein-6. PIC refers to polyinosinic: polycytidylic acid. ODN refers to oligodeoxynucleotides. CRT refers to hydrocortisone. SCP refers to stimulated cytokine production, and SI refers to sensitivity to inhibition. N/A refers to not available in the dataset.

Results

Overview of integrated data

We integrated all five of our lab's studies (40–44) that included (a) relevant measures for at least one of the primary predictors socioeconomic disadvantage or chronic psychological stress and (b) in vitro measures of stimulated cytokine production and sensitivity to glucocorticoid inhibition. Of these five studies, three utilized longitudinal designs and had two waves of data about 1.5 to 2 years apart. The integrated dataset resulted in a sample of 1,607 individuals (960 with longitudinal data), diverse in age (8 to 64 years old) and race (53% White [n = 857], 22% Asian [n = 359], 13% Black [n = 205], and 9% Latino/a [n = 151]). Table 1 summarizes sample and demographic characteristics by study. Supplementary Materials provide full descriptions of methods and statistical approach; below, we provide overall descriptions of the measures.

Socioeconomic disadvantage was assessed with family gross income, family savings, and education (parental education for youth). To harmonize family income and family savings, dollars that were reported in Canadian Dollars were first converted to US Dollars using conversion rates at the time of data collection, and then uniformly recoded into a 9-point scale. Education was similarly harmonized by recoding into a uniform 5-point scale. Finally, a composite was created by averaging the inverse of the standardized scores of these measures, such that higher values indicate greater disadvantage.

Chronic psychological stress was assessed with ageappropriate versions of the UCLA Life Stress Interview (45), which is a semistructured interview focusing on stress over the past 6 months experienced across multiple life domains, including family relationships, friendships, and either school (for youth) or work (for adults). We opted for this approach, rather than self-reported questionnaires, because trained interviewers made developmentally appropriate and contextually informed ratings of chronic stress. As such, this approach allows integration of data across studies despite the wide age range and resulting vast variations in life experiences. To simplify analyses, and capture each participant's overall chronic stress burden, we used a composite measure averaging ratings across the life domains. For both socioeconomic disadvantage and chronic stress, we used ratings made at baseline in studies with longitudinal designs to allow prospective analyses for the most rigorous examination of hypotheses.

Proinflammatory phenotype was assessed with two components of the inflammatory process: stimulated production of cytokines following a microbial challenge, and sensitivity to inhibitory signals that typically regulate this response. To assess stimulated cytokine production, antecubital blood was collected, and cells were dispensed into wells with one or more activating ligand(s) that stimulated the production of proinflammatory cytokines. Supernatants were harvested after incubation, and proinflammatory cytokines were measured via immunoassay or flow cytometry. Proinflammatory cytokines included IL-1 β , IL-6, IL-8, IL-10, and TNF- α , but studies varied in the specific panel of cytokines assayed, which is summarized in Table 1. Composites were formed by averaging across ligands and cytokines. In all studies, higher values indicate greater stimulated cytokine production.

To assess sensitivity to inhibitory signals, cells were coincubated with the activating ligand and varying doses of hydrocortisone, a molecule that provides anti-inflammatory feedback. Each study quantified the same panel of proinflammatory cytokines as above, and all coefficients of variations were less than 10%. Multilevel modeling was used to estimate and extract within-individual slopes relating dose of inhibitor and amount of cytokine production (46). These slopes had negative signs meaning that as the inhibitor dose increased, cytokine production decreased. To aid interpretation, we subsequently inversed slope values, such that higher values indicate greater sensitivity to inhibition signals. As above, a composite was created for each study by averaging the standardized slopes across cytokine. Any deviations from these general approaches are detailed in the Supplementary Materials. Immune measures were assessed both at baseline and at the follow-up visit for longitudinal studies.

To harmonize these immune measures and to define a personfocused inflammatory phenotype, latent profile analyses were conducted within each study using the readouts described above: stimulated cytokine production and sensitivity to inhibition signals. We fixed the number of profiles to two because doing so allowed us to harmonize measures across studies despite the heterogenous methods (summarized in Table 1). Moreover, across the five studies and across timepoints (for longitudinal studies), the two profiles exhibited the same hypothesized pattern of characteristics. Specifically, as summarized in Table S1 and depicted in Figure S1, one profile is characterized by relatively higher stimulated cytokine production and relatively lower sensitivity to inhibition signals (referred to as the "proinflammatory phenotype"), whereas the other profile is characterized by relatively low stimulated cytokine production and relatively high sensitivity to inhibition signals. The profiles were dummy-coded with the proinflammatory phenotype coded as 1. Some temporal stability in membership in the proinflammatory phenotype profile was demonstrated across the 1.5 to 2 years, OR = 13.23 (r equivalent = 0.51).

We also assessed age, sex at birth (male vs. female), race (dummy-coded with White as reference: Asian, Black, Latino/a, and other people of color), and adiposity (body mass index) as covariates, which were entered in all models. We controlled for race because there are known racial disparities in SES, chronic stress, and inflammation. We controlled for race because there are known racial disparities in SES, chronic stress, and inflammation. However, it is important to note that these racial disparities should not be interpreted as reflecting fixed or innate biological differences; rather, multiple sources indicate that they are products of experiences, policies, and practices that have systemically disadvantaged subgroups of individuals based on skin color (47–49). As controlling for these systemic influences is infeasible with our datasets, we used race as a covariate to proxy for them.

Preliminary analyses

Bivariate associations

Relative to White participants, Black and Latino/a participants had greater socioeconomic disadvantage (r = 0.21 and r = 0.09, respectively), and Asian participants had less socioeconomic disadvantage (r = -0.08). Black participants also had higher adiposity (r = 0.07) and Asian participants had lower adiposity (r = -0.09) compared to White participants. No other race or ethnicity dif-

ferences emerged. Increased age was associated with greater adiposity (r = 0.30), greater socioeconomic disadvantage (r = 0.14), and more chronic stress (r = 0.29). No other age-dependent associations emerged. Female (vs. male) participants had greater socioeconomic disadvantage (r = 0.06). No other sex differences emerged. Finally, as would be expected, more socioeconomic disadvantage was associated with higher chronic stress (r = 0.38).

Main associations of socioeconomic disadvantage and chronic stress

We then conducted logistic regressions to examine the links from socioeconomic disadvantage and chronic psychological stress to proinflammatory phenotype profile. As summarized in Table S2 and depicted in Fig. 1(A), individuals experiencing greater socioeconomic disadvantage were more likely to have cells that exhibited the proinflammatory phenotype (b = 0.18, SE = 0.06, P = 0.001). Specifically, every SD increase in disadvantage was associated with 20% higher odds of having the proinflammatory phenotype, following adjustment for age, sex, race, and BMI. Similarly, as depicted in Fig. 1(B), chronic stress was also associated with the proinflammatory phenotype (b = 0.29, SE = 0.06, P < 0.001), such that every SD increase in chronic stress rating was associated with 33% higher odds of having the proinflammatory phenotype following covariate adjustment.

Next, we examined these associations longitudinally. As summarized in Table S2 and depicted in Fig. 1(C) and (D), controlling for covariates, both socioeconomic disadvantage and chronic stress were associated with proinflammatory phenotype at Time 2 (b = 0.20, SE = 0.08, P = 0.009 and b = 0.34, SE = 0.08, P < 0.001, respectively). Specifically, every SD increase in disadvantage and chronic stress were associated with 23% and 41% increased odds of exhibiting the proinflammatory phenotype 1.5 to 2 years later, following covariate adjustment. We also did change analyses, focusing on change in proinflammatory phenotype at Time 1 to Time 2. Controlling for proinflammatory phenotype at Time 1 and the same panel of covariates, chronic stress, but not disadvantage, remained associated with proinflammatory phenotype at Time 2 (b = 0.28, SE = 0.09, P = 0.001).

Hypothesis testing Indirect effects

First, we conducted path models to examine whether chronic psychological stress accounted for the cross-sectional and longitudinal associations between socioeconomic disadvantage and proinflammatory phenotype. As depicted in Fig. 2(A), there was a significant indirect effect cross-sectionally (b = 0.08, SE = 0.02, CI₉₅ = [.04, 0.12]), such that disadvantage was associated with higher chronic stress (b = 0.32, SE = 0.02, CI₉₅ [.27, 0.37]), which in turn was associated with increased odds of exhibiting a proinflammatory phenotype (OR = 1.28, b = 0.25, SE = 0.06, CI₉₅ [.13, 0.38]), following adjustment for age, sex at birth, race, and adiposity.

Similarly, as depicted in Fig. 2(B) and (C), there were also significant indirect effects longitudinally, (b = 0.09, SE = 0.03, CI₉₅ = [.04, 0.15]), and in terms of prospective change from Time 1 to Time 2 (b = 0.08, SE = 0.03, CI₉₅ = [.03, 0.15]). In both cases, controlling for covariates, more socioeconomic disadvantage at Time 1 was associated with higher chronic stress (longitudinal: b = 0.31, SE = 0.03, CI₉₅ [.24, 0.37]; prospective change: b = 0.30, SE = 0.03, CI₉₅ [.23, 0.36]), which in turn was associated with increased odds of exhibiting a proinflammatory phenotype at Time 2 (longitudinal: OR = 1.35, b = 0.30, SE = 0.08, CI₉₅ [.14, 0.46], and with increases from Time 1 to Time 2 in the odds of exhibiting a



Fig. 1. Associations from socioeconomic disadvantage and chronic stress to proinflammatory phenotype cross-sectionally (Panels A and B), longitudinally (Panels C and D). Boxplots are shown with means displayed in diamonds. Raw datapoints are displayed. Note that although some datapoints appear to be outliers, they did not pass a-priori outlier threshold of +/- 3SD from the mean; nonetheless, when winsorized, all results remained the same.

proinflammatory phenotype: OR = 1.32, b = 0.28, SE = 0.09, CI_{95} [.10, 0.46]). Finally, to examine directionality, we examined an indirect effect that is temporally reversed than hypothesized; that is, whether chronic stress at Time 2 would account for the link between disadvantage at Time 2 and proinflammatory phenotype at Time 1. We found no evidence for such indirect effect (b = 0.000, SE = 0.032, CI_{95} [-0.06, 0.07]).

Sensitive period hypothesis

Next, we examined whether the indirect effects from disadvantage to chronic stress to proinflammatory phenotype varied by age. They did in all three cases, as reflected in significant moderated indirect effects for cross-sectional analysis (b = -0.004, SE = 0.001, CI₉₅[-0.007, -0.002]), longitudinal analysis (b = -0.004, SE = 0.002, CI₉₅[-0.008, -0.001]), and in prospective change analysis (b = -0.004, SE = 0.002, CI₉₅[-0.008, -0.001]). As depicted in Fig. 3(A), in all three cases, the simple indirect effects from disadvantage to proinflammatory phenotype via chronic stress were strongest at the youngest ages and linearly declined across the lifecourse.

Further decompositions revealed that these age-moderated indirect effects were driven by age moderation of the path between chronic stress and proinflammatory phenotype. Specifically, as depicted in Fig. 3(B), the magnitude of the link between chronic stress and proinflammatory phenotype was strongest at younger ages and declined across the lifecourse for cross-sectional analysis (age 10: OR = 2.08, age 20: OR = 1.73, age 30: OR = 1.46, age 40: OR = 1.22, age 50: OR = 1.02), longitudinal analysis (age 10: OR = 2.41, age 20: OR = 1.93, age 30: OR = 1.54, age 40: OR = 1.23, age 50 OR = 0.98), and prospective change analysis (age 10: OR = 2.41, age 20: OR = 1.93, age 30: OR = 1.54, age 40: OR = 1.23, age 50 OR = 0.98). Moreover, region of significance analyses suggested that these age moderations became nonsignificant at age 38 for cross-sectional analysis, age 36 for longitudinal analysis, and age 33 for prospective change analysis. In other words, the link between chronic stress and proinflammatory phenotype was observed from childhood well into early adulthood, but not after.

Finally, because there were more youth studies than adult studies, age was not evenly distributed; therefore, we conducted sensitivity analyses with age recoded into three developmental stages, each with adequate sample sizes (50, 51): childhood (under age 14; N = 372), adolescence (ages 14 to 18; N = 618), adulthood (age 19 and above; N = 634). Comparable results were observed such that all three moderated indirect effects remained significant.



Fig. 2. Indirect effect models wherein socioeconomic disadvantage was associated with greater chronic psychological stress, which in turn was associated with proinflammatory phenotype cross-sectionally (Panel A), longitudinally (Panel B), and in terms of prospective change (Panel C). Coefficients were adjusted for race, sex at birth, and body mass index. Prospective change indirect effect model additionally controlled for proinflammatory phenotype at Time 1. In all the three panels, socioeconomic disadvantage and chronic psychological stress were assessed at Time 1.

Discussion

This integrated data analysis aimed to address two unresolved questions. First, is socioeconomic disadvantage linked to chronic psychological stress, which in turn relates to the expression of a proinflammatory phenotype? We found evidence for such indirect effects cross-sectionally, longitudinally, and in terms of prospective change over time. These findings are consistent with conceptual models specifying that disadvantage can give rise to many chronic psychological stressors, including strained relationships and academic or vocational difficulties, which in turn, can alter the operating tendencies of immune cells toward mounting more aggressive inflammatory responses upon challenge and being less sensitive to inhibition signals (31). Of note, we found no evidence for a temporally reversed scenario linking disadvantage and chronic stress at Time 2 to proinflammatory phenotypes at Time 1, shedding light on the directionality of associations.

Second, is there a developmental period during which inflammatory responses are more sensitive to environmental input? Results suggest that the magnitude of the indirect associations from socioeconomic disadvantage to proinflammatory phenotype via chronic stress were strongest in earlier decades of life and weakened across the lifecourse. These findings are consistent with the conceptual notion of a sensitive period early in life during which bodily tissues are rapidly developing and thus have heightened plasticity to environmental inputs (e.g. 30 to 33). Of note, our findings suggest that despite attenuation, plasticity is retained into early adulthood (mid-30's), suggesting a more prolonged capacity for environmental calibration of physiology than traditionally theorized (52), and highlighting the importance of a lifecourse approach in both conceptual and empirical examinations.

The current findings underscore the utility of a lifecourse perspective to conceptual models of stress. While this study begins to address lifecourse queries, it also invites additional temporal questions about whether there are conditions under which stress-inflammation links would exhibit sensitive period patterns, with associations larger at younger ages, versus accumulation patterns, with associations largest at older ages. One factor may be how inflammatory activity is assessed. Specifically, the stimulation and inhibition measures used here are functional in nature, and therefore, might show the greatest plasticity to experience during sensitive periods of development, when the immune system's operating tendencies are being calibrated. By contrast, circulating inflammatory biomarkers, such as CRP, are relatively static indicators that are thought to proxy unresolved inflammation. Elevations in these biomarkers take time to accumulate, and their association with stress might, therefore, be expected to strengthen with age. Indeed, in other investigations, we have demonstrated that socioeconomic disparities in circulating inflammation strengthens with age using an integrative data analysis approach (53) and a meta-analytical approach (54). These findings point to the importance of considering both the developmental timing of stress exposure and the nature of the inflammatory parameter being considered. These findings also have practical implications for efforts to mitigate health disparities. Specifically, both the patterns from the present study of mechanistic inflammatory processes and the patterns from our previous investigations of circulating inflammation suggest that prevention efforts may be most efficacious when implemented during childhood. However, as the current findings also observed malleability well into early adulthood, there is still utility in implementing interventions during adolescence and adulthood.

The strengths of this integrated data analysis include the use of a large, pooled sample of over 1,500 participants that is demographically diverse, increasing the generalizability of findings. This was achieved without the use of public data, for which there are increasing concerns about their overuse, creating dependencies among research papers, giving false impressions of independent contributions, and exacerbating sample-specific peculiarities that may not replicate (20). In addition to its aggregated size,



Fig. 3. Indirect effects from disadvantage to chronic stress to proinflammatory phenotype moderated by age (Panel A). Higher values on the x-axis indicate larger indirect effect estimates with horizontal bars indicating 95% CI, and higher values on the y-axis indicate older age. The link between chronic psychological stress and proinflammatory phenotype moderated by age (Panel B). Higher values on the x-axis indicate more chronic psychological stress, and higher values on the y-axis indicate higher estimated probability of being in proinflammatory phenotype with vertical bars indicating 95% CI.

the samples in this investigation are also unique in the richness of their data, including interview data on socioeconomic disadvantage and chronic psychological stress as well as in vitro cell culture data to assess functional inflammatory processes. Finally, by incorporating longitudinal data, we were also able to assess changes, shedding light on the temporal precedence of social factors.

There are also several limitations to this analysis. First, there was considerable methodological variability across studies in the cell-culture data, making it difficult to harmonize these measures across studies. We attempted to overcome this challenge via the use of latent profile analyses to estimate a proinflammatory phenotype, which provided the benefit of revealing coupling patterns of these measures. However, to harmonize across studies, we fixed the number of profiles at two. Future research is necessary to explore the optimal number of profiles. Second, relative to circulating biomarkers of inflammation, the clinical predictive value of the functional measures of inflammation used here is less wellestablished. Clarifying the prognostic significance of these assays will also be an important task in subsequent research. Third, an integrative analysis of longitudinal data allowed the opportunity to efficiently examine lifecourse questions; however, this approach is limited by the data available across all studies as well as the duration at which measures were repeated. Here, we did

not have data on early life stress for adult participants and the duration between Time 1 and Time 2 (about 1.5 to 2 years) is relatively short for lifecourse inquiries. As such, it is possible that the observed age effects on the stress-inflammation link is due to participants having developed an adaptation to higher levels of cytokine and, thus also developed lowered sensitivity to inhibition as they aged. Indeed, we observed that older participants had decreased odds of exhibiting the proinflammatory phenotype; however, all models controlled for main effects of age, suggesting that the observed interaction between age and stress were independent of this general age trend. An ideal design would be a "cradle-to-grave" study that follows multiple cohorts from birth to death, repeating assessments of stress and proinflammatory phenotype across the lifespan, and then examining whether the within-individual link between stress and inflammation is moderated by timing of stress exposure. Fourth, whole blood, rather than isolated monocytes, were used in cell culture experiments. Previous research has compared LPS-evoked cytokine production in cultures of whole blood and monocytes, and found correspondence between these methods, concluding that whole blood culture is a valid method for assessing monocytic cytokine responsivity (55). Still, relative to isolated monocytes, the whole blood approach may have introduced additional noise into our results. Furthermore, as chronic stress has been found to promote mobilization of monocytes from the spleen and bone marrow (56), higher stress-related values may partially reflect variations in the number or fraction of monocytes present in culture, rather than per-cell alterations to monocytes' operating tendencies. Future research can help clarify this by stimulating isolated monocyte populations or conducting flow cytometric analyses of monocytespecific cytokine production.

To conclude, the link between stress and inflammation is wellestablished, but most extant studies have relied on circulating markers of inflammation and do not examine how associations may vary across the lifecourse. This integrated data analysis found that socioeconomic disadvantage operated through chronic psychological stress to confer risks for developing a proinflammatory phenotype characterized by cells mounting more pronounced inflammatory responses and being less sensitive to inhibition signals that typically terminate their responses. Of note, the magnitude of these associations was strongest in the early decades of life and declined with age, suggesting a sensitive developmental period during which immune functioning may be most sensitive to environmental calibration.

Methods

This integrative data analysis combined all five of our lab's studies that have relevant data (40–44). Studies were approved by the institutional review board of the university where they were conducted. Adults and a guardian of youth gave consent, and youth gave assent. Supplementary materials provide full description of the method and analytical approach. Data and analysis codes available on Github repository: https://github.com/phoebehlam/ mega2.

Supplementary Material

Supplementary material is available at PNAS Nexus online.

Funding

This research was supported by grants from the National Institutes of Health R01 HD058502 (GEM), R01 HL122328 (GEM), R01 HL108723 (EC), R01 HD093718 (EC), and F31 HL147509 (PHL), grants from the Canadian Institutes of Health Research 67191 (GEM) and 97872 (EC), and a Grant-In-Aid from the American Heart Association (GEM).

Authors' Contributions

E.C. and G.E.M. designed and collected data for all studies; P.H.L. conducted the analyses; and E.C., G.E.M., J.J.C., and P.H.L. wrote and/or revised the manuscript.

References

- 1. Dong M et al. 2004. Insights into causal pathways for ischemic heart disease. Circulation. 110:1761–1766.
- 2. Dube SR et al. 2009. Cumulative childhood stress and autoimmune diseases in adults. Psychosom Med. 71:243–250.
- Felitti VJ et al. 1998. Relationship of childhood abuse and household dysfunction to many of the leading causes of death in adults: the adverse childhood experiences (ACE) study. Am J Prevent Med. 14:245–258.
- 4. Galobardes B, Smith GD, Lynch JW. 2006. Systematic review of the influence of childhood socioeconomic circumstances on risk

for cardiovascular disease in adulthood. Ann Epidemiol. 16:91–104.

- Kittleson MM et al. 2006. Association of childhood socioeconomic status with subsequent coronary heart disease in physicians. Arch Int Med. 166:2356–2361.
- Wegman HL, Stetler C. 2009. A meta-analytic review of the effects of childhood abuse on medical outcomes in adulthood. Psychosom Med. 71:805–812.
- Nathan C, Ding A. 2010. Nonresolving inflammation. Cell. 140:871–882.
- Danesh J et al. 2000. Low grade inflammation and coronary heart disease: prospective study and updated meta-analyses. BMJ. 321:199–204.
- Danesh J et al. 2008. Long-term interleukin-6 levels and subsequent risk of coronary heart disease: two new prospective studies and a systematic review. PLoS Med. 5:e78.
- Pearson TA et al. 2003. Markers of inflammation and cardiovascular disease: application to clinical and public health practice: a statement for healthcare professionals from the Centers for Disease Control and Prevention and the American Heart Association. Circulation. 107:499–511.
- Liu RS et al. 2017. Socioeconomic status in childhood and C reactive protein in adulthood: a systematic review and metaanalysis. J Epidemiol Commun Health. 71:817–826.
- Milaniak I, Jaffee SR. 2019. Childhood socioeconomic status and inflammation: a systematic review and meta-analysis. Brain Behav Immun. 78:161–176.
- Muscatell KA, Brosso SN, Humphreys KL. 2018. Socioeconomic status and inflammation: a meta-analysis. Mol Psych. 1:2189– 2199.
- Armutcu F. 2019. Organ crosstalk: the potent roles of inflammation and fibrotic changes in the course of organ interactions. Inflam Res. 68:825–839.
- Lackey DE, Olefsky JM. 2016. Regulation of metabolism by the innate immune system. Nat Rev Endocrinol. 12:15–28.
- Man K, Kutyavin VI, Chawla A. 2017. Tissue immunometabolism: development, physiology, and pathobiology. Cell Metabol. 25:11– 26.
- 17. Hunter CA, Jones SA. 2015. IL-6 as a keystone cytokine in health and disease. Nat Immunol. 16:448–457.
- Chen E et al. 2006. Socioeconomic status and inflammatory processes in childhood asthma: the role of psychological stress. J Aller Clin Immunol. 117:1014–1020.
- Jiang Y et al. 2021. Socioeconomic status, financial stress, and glucocorticoid resistance among youth with asthma: testing the moderation effects of maternal involvement and warmth. Brain Behav Immun. 96:92–99.
- Mroczek DK, Weston SJ, Graham EK, Willroth EC. 2021. Data overuse in aging research: emerging issues and potential solutions. Psychol Aging. 37:141–147.
- Schreier H, Chen E. 2013. Socioeconomic status and the health of youth: a multilevel, multidomain approach to conceptualizing pathways. Psychol Bull. 139:606.
- 22. Braveman P, Barclay C. 2009. Health disparities beginning in childhood: a life-course perspective. Pediatrics. 124:S163–S175.
- 23. Pampel FC, Krueger PM, Denney JT. 2010. Socioeconomic disparities in health behaviors. Annu Rev Sociol. 36:349–370.
- Chen E, Miller GE. 2013. Socioeconomic status and health: mediating and moderating factors. Annu Rev Clin Psychol. 9: 723–749.
- Bird ST, Bogart LM. 2001. Perceived race-based and socioeconomic status(SES)-based discrimination in interactions with health care providers. Ethn Dis. 11:554–563.

- 26. Steptoe A, Feldman PJ. 2001. Neighborhood problems as sources of chronic stress: development of a measure of neighborhood problems, and associations with socioeconomic status and health. Ann Behav Med. 23:177–185.
- 27. Foster H, Brooks-Gunn J, Martin A. 2007. Poverty/socioeconomic status and exposure to violence in the lives of children and ado-lescents. Cambridge: Cambridge University Press.
- Christian LM, Kowalsky JM, Mitchell AM, Porter K. 2018. Associations of postpartum sleep, stress, and depressive symptoms with LPS-stimulated cytokine production among African American and White women. J Neuroimmunol. 316:98–106.
- 29. Davis MC *et al.* 2008. Chronic stress and regulation of cellular markers of inflammation in rheumatoid arthritis: implications for fatigue. Brain Behav Immun. 22:24–32.
- Barker DJ. 1993. Fetal origins of coronary heart disease. Brit Heart J. 69:195.
- Miller GE, Chen E, Parker KJ. 2011. Psychological stress in childhood and susceptibility to the chronic diseases of aging: moving toward a model of behavioral and biological mechanisms. Psychol Bull. 137:959.
- Nusslock R, Miller GE. 2016. Early-life adversity and physical and emotional health across the lifespan: a neuroimmune network hypothesis. Biol Psych. 80:23–32.
- Tottenham N. 2014. The importance of early experiences for neuro-affective development. Curr Top Behav Neurosci. 16:109– 129.
- 34. Liu D et al. 1997. Maternal care, hippocampal glucocorticoid receptors, and hypothalamic-pituitary-adrenal responses to stress. Science. 277:1659–1662.
- Meaney MJ. 2001. Maternal care, gene expression, and the transmission of individual differences in stress reactivity across generations. Ann Rev Neurosci. 24:1161–1192.
- Meaney MJ, Aitken DH. 1985. The effects of early postnatal handling on hippocampal glucocorticoid receptor concentrations: temporal parameters. Dev Brain Res. 22:301–304.
- Simon AK, Hollander GA, McMichael A. 2015. Evolution of the immune system in humans from infancy to old age. Proc R Soc B Biol Sci. 282:20143085.
- Nikolich-Žugich J. 2018. The twilight of immunity: emerging concepts in aging of the immune system. Nat Immunol. 19:10– 19.
- Hawkley LC, Cacioppo JT. 2004. Stress and the aging immune system. Brain Behav Immun. 18:114–119.
- 40. Miller GE *et al.* 2018. Functional connectivity in central executive network protects youth against cardiometabolic risks linked with neighborhood violence. Proc Natl Acad Sci. 115:12063–12068.

- Chen E, Lee WK, Cavey L, Ho A. 2013. Role models and the psychological characteristics that buffer low-socioeconomic-status youth from cardiovascular risk. Child Dev. 84:1241–1252.
- Miller GE, Cole SW. 2012. Clustering of depression and inflammation in adolescents previously exposed to childhood adversity. Biol Psych. 72:34–40.
- Hostinar CE, Ross KM, Chen E, Miller GE. 2017. Early-life socioeconomic disadvantage and metabolic health disparities. Psychosomatic Med. 79:514.
- 44. Chen E et al. 2017. Difficult family relationships, residential greenspace, and childhood asthma. Pediatrics. 139:e20163056.
- Hammen C et al. 1987. Maternal affective disorders, illness, and stress: risk for children's psychopathology. Am J Psych. 144:736– 741.
- Chiang JJ et al. 2019. Familism and inflammatory processes in African American, Latino, and White youth. Health Psychol. 38:306.
- 47. Gravlee CC. 2009. How race becomes biology: embodiment of social inequality. Am J Phys Anthropol. 139:47–57.
- Kuzawa CW, Gravlee CC. 2016. Beyond genetic race: biocultural insights into the causes of racial health disparities. In: New directions in biocultural anthropology. New York (NY): Wiley, 89– 105.
- Bryant BE, Jordan A, Clark US. 2022. Race as a social construct in psychiatry research and practice. JAMA Psych. 79:93–94.
- Repetti RL, Robles TF, Reynolds B. 2011. Allostatic processes in the family. Dev Psychopathol. 23:921–938.
- 51. WHO. 2021. Adolescent health. [accessed 2021 Jul 14].
- Bruer JT. 1999. The myth of the first three years: a new understanding of early brain development and lifelong learning. New York (NY): Simon and Schuster.
- Lam PH, Chiang JJ, Chen E, Miller GE. 2021. Race, socioeconomic status, and low-grade inflammatory biomarkers across the lifecourse: a pooled analysis of seven studies. Psychoneuroendocrinology. 123:104917.
- Chiang J, Lam P, Chen E, Miller G. in press. Psychological stress during childhood and adolescence and its association with inflammation across the lifespan: a critical review and metaanalysis. Psychol Bull.148:27–66.
- 55. Damsgaard CT, Lauritzen L, Calder PC, Kjær TM, Frøkiær H. 2009. Whole-blood culture is a valid low-cost method to measure monocytic cytokines—a comparison of cytokine production in cultures of human whole-blood, mononuclear cells and monocytes. J Immunol Methods. 340:95–101.
- Weber MD, Godbout JP, Sheridan JF. 2017. Repeated social defeat, neuroinflammation, and behavior: monocytes carry the signal. Neuropsychopharmacology. 42:46–61.